Author's response to reviews

Title: Smoking and mental illness: results from population surveys in Australia and the United States

Authors:

David Lawrence (D.Lawrence@curtin.edu.au)
Francis Mitrou (francism@ichr.uwa.edu.au)
Stephen R Zubrick (S.Zubrick@curtin.edu.au)

Version: 4 Date: 5 August 2009

Author's response to reviews:

The following changes have been made to the manuscript and figures as requested:

Major revisions

Reference 12 ¿ please provide full and accurate details for this reference. This means for an Article providing year/volume/issue/page numbers and for a book the publisher/city/year.

The reference has been updated.

Table in parts - Table 4 has been divided into parts, a and b. It is important that tables are numbered in ascending numerical order - 1, 2, 3, 4 & 5. Can we please ask that the tables are renumbered and any references to the tables in the text are updated to reflect these changes? Alternatively, please merge the tables into one larger complete table.

The two pages of table four have been merged into a single table.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change `The Title of My Prestigious Publication: A Scientific Study¿. To `The title of my prestigious publication: a scientific study¿

The title is in sentence case. The words 'Australia' 'United' and 'States' are appropriately capitalised.

'Email addresses'

Please remove this subheading from the title page
This subheading was included based on your Microsoft word template, but has been removed.

Table title and legend - The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.

Table titles appear above each table, table legends and keys are below each table.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

The figures have been cropped as requested.